Author's response to reviews

Title: Preferred intensity exercise for adolescents receiving treatment for depression: A pragmatic randomised controlled trial

Authors:

Tim Carter (timothy.carter@nottingham.ac.uk)
Boliang Guo (guo.boliang@nottingham.ac.uk)
David Turner (David.A.Turner@uea.ac.uk)
Iaonnis Morres (iomorres@pe.uth.gr)
Elizabeth Khalil (Elizabeth.khalil@nottingham.ac.uk)
Emily Brighton (emilybrighton@hotmail.co.uk)
Marie Armstrong (marie.armstrong@nottshc.nhs.uk)
Patrick Callaghan (patrick.callaghan@nottingham.ac.uk)

Version: 4 Date: 21 September 2015

Author's response to reviews: see over
Dear editor

Thank you for your feedback. Please find a point-by-point response to the concerns below:

1. Abbreviation section has been added
2. Figure legends have been changed as stated
3. Aims have been subsumed into the background section in the abstract as directed

Kind regards

Tim carter